
%% bare_jrnl.tex
%% V1.4b
%% 2015/08/26
%% by Michael Shell
%% see http://www.michaelshell.org/
%% for current contact information.
%%
%% This is a skeleton file demonstrating the use of IEEEtran.cls
%% (requires IEEEtran.cls version 1.8b or later) with an IEEE
%% journal paper.
%%
%% Support sites:
%% http://www.michaelshell.org/tex/ieeetran/
%% http://www.ctan.org/pkg/ieeetran
%% and
%% http://www.ieee.org/

%%*************************************************************************
%% Legal Notice:
%% This code is offered as-is without any warranty either expressed or
%% implied; without even the implied warranty of MERCHANTABILITY or
%% FITNESS FOR A PARTICULAR PURPOSE! 
%% User assumes all risk.
%% In no event shall the IEEE or any contributor to this code be liable for
%% any damages or losses, including, but not limited to, incidental,
%% consequential, or any other damages, resulting from the use or misuse
%% of any information contained here.
%%
%% All comments are the opinions of their respective authors and are not
%% necessarily endorsed by the IEEE.
%%
%% This work is distributed under the LaTeX Project Public License (LPPL)
%% ( http://www.latex-project.org/ ) version 1.3, and may be freely used,
%% distributed and modified. A copy of the LPPL, version 1.3, is included
%% in the base LaTeX documentation of all distributions of LaTeX released
%% 2003/12/01 or later.
%% Retain all contribution notices and credits.
%% ** Modified files should be clearly indicated as such, including  **
%% ** renaming them and changing author support contact information. **
%%*************************************************************************


% *** Authors should verify (and, if needed, correct) their LaTeX system  ***
% *** with the testflow diagnostic prior to trusting their LaTeX platform ***
% *** with production work. The IEEE's font choices and paper sizes can   ***
% *** trigger bugs that do not appear when using other class files.       ***                          ***
% The testflow support page is at:
% http://www.michaelshell.org/tex/testflow/



\documentclass[journal]{IEEEtran}
%
% If IEEEtran.cls has not been installed into the LaTeX system files,
% manually specify the path to it like:
% \documentclass[journal]{../sty/IEEEtran}





% Some very useful LaTeX packages include:
% (uncomment the ones you want to load)


% *** MISC UTILITY PACKAGES ***
%
%\usepackage{ifpdf}
% Heiko Oberdiek's ifpdf.sty is very useful if you need conditional
% compilation based on whether the output is pdf or dvi.
% usage:
% \ifpdf
%   % pdf code
% \else
%   % dvi code
% \fi
% The latest version of ifpdf.sty can be obtained from:
% http://www.ctan.org/pkg/ifpdf
% Also, note that IEEEtran.cls V1.7 and later provides a builtin
% \ifCLASSINFOpdf conditional that works the same way.
% When switching from latex to pdflatex and vice-versa, the compiler may
% have to be run twice to clear warning/error messages.






% *** CITATION PACKAGES ***
%
%\usepackage{cite}
% cite.sty was written by Donald Arseneau
% V1.6 and later of IEEEtran pre-defines the format of the cite.sty package
% \cite{} output to follow that of the IEEE. Loading the cite package will
% result in citation numbers being automatically sorted and properly
% "compressed/ranged". e.g., [1], [9], [2], [7], [5], [6] without using
% cite.sty will become [1], [2], [5]--[7], [9] using cite.sty. cite.sty's
% \cite will automatically add leading space, if needed. Use cite.sty's
% noadjust option (cite.sty V3.8 and later) if you want to turn this off
% such as if a citation ever needs to be enclosed in parenthesis.
% cite.sty is already installed on most LaTeX systems. Be sure and use
% version 5.0 (2009-03-20) and later if using hyperref.sty.
% The latest version can be obtained at:
% http://www.ctan.org/pkg/cite
% The documentation is contained in the cite.sty file itself.




% *** GRAPHICS RELATED PACKAGES ***
%
\ifCLASSINFOpdf
  \usepackage[pdftex]{graphicx}
  % declare the path(s) where your graphic files are
  % \graphicspath{{../pdf/}{../jpeg/}}
  % and their extensions so you won't have to specify these with
  % every instance of \includegraphics
  % \DeclareGraphicsExtensions{.pdf,.jpeg,.png}
\else
  % or other class option (dvipsone, dvipdf, if not using dvips). graphicx
  % will default to the driver specified in the system graphics.cfg if no
  % driver is specified.
  %\usepackage[dvips]{graphicx}
  \usepackage[dvipdfmx]{graphicx}
  % declare the path(s) where your graphic files are
  % \graphicspath{{../eps/}}
  % and their extensions so you won't have to specify these with
  % every instance of \includegraphics
  % \DeclareGraphicsExtensions{.eps}
\fi
% graphicx was written by David Carlisle and Sebastian Rahtz. It is
% required if you want graphics, photos, etc. graphicx.sty is already
% installed on most LaTeX systems. The latest version and documentation
% can be obtained at: 
% http://www.ctan.org/pkg/graphicx
% Another good source of documentation is "Using Imported Graphics in
% LaTeX2e" by Keith Reckdahl which can be found at:
% http://www.ctan.org/pkg/epslatex
%
% latex, and pdflatex in dvi mode, support graphics in encapsulated
% postscript (.eps) format. pdflatex in pdf mode supports graphics
% in .pdf, .jpeg, .png and .mps (metapost) formats. Users should ensure
% that all non-photo figures use a vector format (.eps, .pdf, .mps) and
% not a bitmapped formats (.jpeg, .png). The IEEE frowns on bitmapped formats
% which can result in "jaggedy"/blurry rendering of lines and letters as
% well as large increases in file sizes.
%
% You can find documentation about the pdfTeX application at:
% http://www.tug.org/applications/pdftex





% *** MATH PACKAGES ***
%
\usepackage{amsmath}
% A popular package from the American Mathematical Society that provides
% many useful and powerful commands for dealing with mathematics.
%
% Note that the amsmath package sets \interdisplaylinepenalty to 10000
% thus preventing page breaks from occurring within multiline equations. Use:
%\interdisplaylinepenalty=2500
% after loading amsmath to restore such page breaks as IEEEtran.cls normally
% does. amsmath.sty is already installed on most LaTeX systems. The latest
% version and documentation can be obtained at:
% http://www.ctan.org/pkg/amsmath





% *** SPECIALIZED LIST PACKAGES ***
%
%\usepackage{algorithmic}
% algorithmic.sty was written by Peter Williams and Rogerio Brito.
% This package provides an algorithmic environment fo describing algorithms.
% You can use the algorithmic environment in-text or within a figure
% environment to provide for a floating algorithm. Do NOT use the algorithm
% floating environment provided by algorithm.sty (by the same authors) or
% algorithm2e.sty (by Christophe Fiorio) as the IEEE does not use dedicated
% algorithm float types and packages that provide these will not provide
% correct IEEE style captions. The latest version and documentation of
% algorithmic.sty can be obtained at:
% http://www.ctan.org/pkg/algorithms
% Also of interest may be the (relatively newer and more customizable)
% algorithmicx.sty package by Szasz Janos:
% http://www.ctan.org/pkg/algorithmicx




% *** ALIGNMENT PACKAGES ***
%
\usepackage{array}
% Frank Mittelbach's and David Carlisle's array.sty patches and improves
% the standard LaTeX2e array and tabular environments to provide better
% appearance and additional user controls. As the default LaTeX2e table
% generation code is lacking to the point of almost being broken with
% respect to the quality of the end results, all users are strongly
% advised to use an enhanced (at the very least that provided by array.sty)
% set of table tools. array.sty is already installed on most systems. The
% latest version and documentation can be obtained at:
% http://www.ctan.org/pkg/array


% IEEEtran contains the IEEEeqnarray family of commands that can be used to
% generate multiline equations as well as matrices, tables, etc., of high
% quality.




% *** SUBFIGURE PACKAGES ***
\ifCLASSOPTIONcompsoc
  \usepackage[caption=false,font=normalsize,labelfont=sf,textfont=sf]{subfig}
\else
  \usepackage[caption=false,font=footnotesize]{subfig}
\fi
% subfig.sty, written by Steven Douglas Cochran, is the modern replacement
% for subfigure.sty, the latter of which is no longer maintained and is
% incompatible with some LaTeX packages including fixltx2e. However,
% subfig.sty requires and automatically loads Axel Sommerfeldt's caption.sty
% which will override IEEEtran.cls' handling of captions and this will result
% in non-IEEE style figure/table captions. To prevent this problem, be sure
% and invoke subfig.sty's "caption=false" package option (available since
% subfig.sty version 1.3, 2005/06/28) as this is will preserve IEEEtran.cls
% handling of captions.
% Note that the Computer Society format requires a larger sans serif font
% than the serif footnote size font used in traditional IEEE formatting
% and thus the need to invoke different subfig.sty package options depending
% on whether compsoc mode has been enabled.
%
% The latest version and documentation of subfig.sty can be obtained at:
% http://www.ctan.org/pkg/subfig




% *** FLOAT PACKAGES ***
%
%\usepackage{fixltx2e}
% fixltx2e, the successor to the earlier fix2col.sty, was written by
% Frank Mittelbach and David Carlisle. This package corrects a few problems
% in the LaTeX2e kernel, the most notable of which is that in current
% LaTeX2e releases, the ordering of single and double column floats is not
% guaranteed to be preserved. Thus, an unpatched LaTeX2e can allow a
% single column figure to be placed prior to an earlier double column
% figure.
% Be aware that LaTeX2e kernels dated 2015 and later have fixltx2e.sty's
% corrections already built into the system in which case a warning will
% be issued if an attempt is made to load fixltx2e.sty as it is no longer
% needed.
% The latest version and documentation can be found at:
% http://www.ctan.org/pkg/fixltx2e


%\usepackage{stfloats}
% stfloats.sty was written by Sigitas Tolusis. This package gives LaTeX2e
% the ability to do double column floats at the bottom of the page as well
% as the top. (e.g., "\begin{figure*}[!b]" is not normally possible in
% LaTeX2e). It also provides a command:
%\fnbelowfloat
% to enable the placement of footnotes below bottom floats (the standard
% LaTeX2e kernel puts them above bottom floats). This is an invasive package
% which rewrites many portions of the LaTeX2e float routines. It may not work
% with other packages that modify the LaTeX2e float routines. The latest
% version and documentation can be obtained at:
% http://www.ctan.org/pkg/stfloats
% Do not use the stfloats baselinefloat ability as the IEEE does not allow
% \baselineskip to stretch. Authors submitting work to the IEEE should note
% that the IEEE rarely uses double column equations and that authors should try
% to avoid such use. Do not be tempted to use the cuted.sty or midfloat.sty
% packages (also by Sigitas Tolusis) as the IEEE does not format its papers in
% such ways.
% Do not attempt to use stfloats with fixltx2e as they are incompatible.
% Instead, use Morten Hogholm'a dblfloatfix which combines the features
% of both fixltx2e and stfloats:
%
% \usepackage{dblfloatfix}
% The latest version can be found at:
% http://www.ctan.org/pkg/dblfloatfix




%\ifCLASSOPTIONcaptionsoff
%  \usepackage[nomarkers]{endfloat}
% \let\MYoriglatexcaption\caption
% \renewcommand{\caption}[2][\relax]{\MYoriglatexcaption[#2]{#2}}
%\fi
% endfloat.sty was written by James Darrell McCauley, Jeff Goldberg and 
% Axel Sommerfeldt. This package may be useful when used in conjunction with 
% IEEEtran.cls'  captionsoff option. Some IEEE journals/societies require that
% submissions have lists of figures/tables at the end of the paper and that
% figures/tables without any captions are placed on a page by themselves at
% the end of the document. If needed, the draftcls IEEEtran class option or
% \CLASSINPUTbaselinestretch interface can be used to increase the line
% spacing as well. Be sure and use the nomarkers option of endfloat to
% prevent endfloat from "marking" where the figures would have been placed
% in the text. The two hack lines of code above are a slight modification of
% that suggested by in the endfloat docs (section 8.4.1) to ensure that
% the full captions always appear in the list of figures/tables - even if
% the user used the short optional argument of \caption[]{}.
% IEEE papers do not typically make use of \caption[]'s optional argument,
% so this should not be an issue. A similar trick can be used to disable
% captions of packages such as subfig.sty that lack options to turn off
% the subcaptions:
% For subfig.sty:
% \let\MYorigsubfloat\subfloat
% \renewcommand{\subfloat}[2][\relax]{\MYorigsubfloat[]{#2}}
% However, the above trick will not work if both optional arguments of
% the \subfloat command are used. Furthermore, there needs to be a
% description of each subfigure *somewhere* and endfloat does not add
% subfigure captions to its list of figures. Thus, the best approach is to
% avoid the use of subfigure captions (many IEEE journals avoid them anyway)
% and instead reference/explain all the subfigures within the main caption.
% The latest version of endfloat.sty and its documentation can obtained at:
% http://www.ctan.org/pkg/endfloat
%
% The IEEEtran \ifCLASSOPTIONcaptionsoff conditional can also be used
% later in the document, say, to conditionally put the References on a 
% page by themselves.




% *** PDF, URL AND HYPERLINK PACKAGES ***
%
\usepackage{url}
% url.sty was written by Donald Arseneau. It provides better support for
% handling and breaking URLs. url.sty is already installed on most LaTeX
% systems. The latest version and documentation can be obtained at:
% http://www.ctan.org/pkg/url
% Basically, \url{my_url_here}.


\usepackage{multirow}
\usepackage[switch]{lineno}
%\linenumbers

% *** Do not adjust lengths that control margins, column widths, etc. ***
% *** Do not use packages that alter fonts (such as pslatex).         ***
% There should be no need to do such things with IEEEtran.cls V1.6 and later.
% (Unless specifically asked to do so by the journal or conference you plan
% to submit to, of course. )


% correct bad hyphenation here
\hyphenation{op-tical net-works semi-conduc-tor}


\begin{document}
\begin{thebibliography}{1}

\bibitem{Shiomi07}
  M. Shiomi, T. Kanda, H. Ishiguro, and N. Hagita,
  ``Interactive humanoid robots for a science museum,''
  \emph{IEEE Intelligent Systems},
  vol. 22,
  no. 2,
  pp. 25--32,
  2007

\bibitem{Triebel16}
  R. Triebel \emph{et al.},
  ``Spencer: a socially aware service robot for passenger guidance and help in busy airports,''
  \emph{Field and Service Robotics},
  vol. 113,
  pp. 607--622,
  2016
  
\bibitem{Gross08}
  H.-M. Gross \emph{et al.},
  ``ShopBot: Progress in developing an interactive mobile shopping assistant for everyday use,''
  \emph{in Proc. IEEE International Conference on Systems, Man and Cybernetics},
  pp. 3471--3478,
  2008

\bibitem{Osawa17}
  H. Osawa \emph{et al.},
  ``Analysis of robot hotel: Reconstruction of works with robots,''
  \emph{in Proc. IEEE International Symposium on Robot and Human Interactive Communication},
  pp. 219--223,
  2017

\bibitem{Yamazaki12}
  K. Yamazaki \emph{et al.},
  ``Home-Assistant Robot for an Aging Society,''
  \emph{in Proc. the IEEE},
  vol. 100,
  no. 8,
  pp. 2429--2441,
  2012

\bibitem{Baba20}
  J. Baba \emph{et al.},
  ``Teleoperated Robot Acting Autonomous for Better Customer Satisfaction,''
  \emph{in Proc. SIGCHI Conference on Human Factors in Computing Systems},
  pp. 1--8,
  2020

\bibitem{Kanda10}
  T. Kanda, M. Shiomi, Z. Miyashita, H. Ishiguro, and N. Hagita,
  ``A Communication Robot in a Shopping Mall,''
  \emph{IEEE Transaction on Robotics},
  vol. 25,
  no. 5,
  pp. 897--913,
  2010

\bibitem{Mende19}
  M. Mende, M. L. Scott, J. van Doorn, D. Grewal, I. Shanks,
  ``Service Robots Rising: How Humanoid Robots Influence Service Experiences and Elicit Compensatory Consumer Responses,''
  \emph{Journal of Marketing Research},
  vol. 56,
  no. 4,
  pp. 535--556,
  2019

\bibitem{Niemela17}
  M. Niemel\"{a}, A. Arvola, and I. Aaltonen,
  ``Monitoring the Acceptance of a Social Service Robot in a Shopping Mall: First Results,''
  \emph{in Proc. ACM/IEEE International Conference on Human-Robot Interaction},
  pp. 225--226,
  2017

\bibitem{Niemela19}
  M. Niemel\"{a}, P. Heikkil\"{a}, H. Lammi, V. Oksman,
  ``A Social Robot in a Shopping Mall: Studies on Acceptance and Stakeholder Expectations,'' in
  \emph{Social Robots: Technological, Societal and Ethical Aspects of Human-Robot Interaction. Human Computer Interaction Series},
  O. Korn, Eds,
  Springer,
  pp. 119--144,
  2019

\bibitem{Lee12}
  M. K. Lee, S. Kiesler, J. Forlizzi, and P. Rybski,
  ``Ripple effects of an embedded social agent: a field study of a social robot in the workplace,''
  \emph{in Proc. SIGCHI Conference on Human Factors in Computing Systems},
  pp. 695--704,
  2012

\bibitem{Tanaka16}
  K. Tanaka, N. Yamashita, H. Nakanishi and H. Ishiguro,
  ``Teleoperated or autonomous?: How to produce a robot operator's pseudo presence in HRI,''
  \emph{in Proc. ACM/IEEE International Conference on Human-Robot Interaction},
  pp. 133--140,
  2016

\bibitem{Okafuji20}
  Y. Okafuji \emph{et al.},
  ``Please Listen to Me: How to Make Passersby Stop by a Humanoid Robot in a Shopping Mall,''
  \emph{in Proc. ACM/IEEE International Conference on Human-Robot Interaction},
  pp. 381--383,
  2020

\bibitem{Shiomi08}
  M. Shiomi \emph{et al.},
  ``A Semi-autonomous Communication Robot - A Field Trial at a Train Station -,''
  \emph{in Proc. ACM/IEEE International Conference on Human-Robot Interaction},
  pp. 303--310,
  2008
  
\bibitem{Shiomi09}
  M. Shiomi \emph{et al.},
  ``Field Trial of Networked Social Robots in a Shopping Mall,''
  \emph{in Proc. International Conference on Intelligent Robots and Systems},
  pp. 2846--2853,
  2009
  
\bibitem{Shi13}
  C. Shi, M. Shiomi, C. Smith, T. Kanda, and H. Ishiguro,
  ``A model of distributional handing interaction for a mobile robot,''
  \emph{in Proc. Robotics: Science and Systems},
  2013

\bibitem{Shi18}
  C. Shi, S. Satake, T. Kanda, and H. Ishiguro,
  ``A Robot that Distributes Flyers to Pedestrians in a Shopping Mall,''
  \emph{International Journal of Social Robotics},
  vol. 10,
  pp. 421--437,
  2018
     
\bibitem{Yamazaki08}
  A. Yamazaki \emph{et al.}, 
  ``Precision timing in human-robot interaction: coordination of head movement and utterance,''
  \emph{in Proc. SIGCHI Conference on Human Factors in Computing Systems},
  pp. 131--140,
  2008

\bibitem{Bergstrom09}
  N. Bergstr\"{o}m, T. Kanda, T. Miyashita, H. Ishiguro, and N. Hagita,
  ``Modeling of Natural Human-Robot Encounters,''
  \emph{in Proc. International Conference on Intelligent Robots and Systems},
  pp. 2623--2629,
  2008
  
\bibitem{Iwasaki18}
  M. Iwasaki, J. Zhou, M. Ikeda, T. Kawamura, and H. Nakanishi,
  ``A Customer's Attitude to a Robotic Salesperson Depends on Their Initial Interaction,''
  \emph{in Proc. IEEE International Symposium on Robot and Human Interactive Communication},
  pp. 300--305,
  2018
  
\bibitem{Iwasaki19}
  M. Iwasaki \emph{et al.},
  ``Acting as if Being Aware of Visitors' Attention Strengthens a Robotic Salesperson's Social Presence,''
  \emph{in Proc. International Conference on Human-Agent Interaction},
  pp. 19--27,
  2019

\bibitem{Shiomi13}
  M. Shiomi \emph{et al.},
  ``Recommendation effects of a social robot for advertisement-use context in a shopping mall,''
  \emph{International Journal of Social Robotics},
  vol. 5,
  no. 2,
  pp. 251--262,
  2013
  
\bibitem{Hayashi07}
  K. Hayashi \emph{et al.},
  ``Humanoid Robots as a Passive-Social Medium - A Filed Experiment at a Train Station -,''
  \emph{in Proc. International Conference on Human-Robot Interaction},
  pp. 137--144,
  2007

\bibitem{Michalowski06}
  M. P. Michalowski, S. Sabanovic, and R. Simmons,
  ``A spatial model of engagement for a social robot,''
  \emph{in Proc. IEEE International Workshop on Advanced Motion Control},
  pp. 762--767,
  2006
  
\bibitem{Li10}
  D. Li, P. L. P. Rau, Y. Li,
  ``A Cross-cultural Study: Effect of Robot Appearance and Task,''
  \textit{International Journal of Social Robot},
  vol. 2,
  pp. 175--186,
  2010

\bibitem{Trovato13}
  G. Trovato \emph{et al.},
  ``Cross-cultural study on human-robot greeting interaction: acceptance and discomfort by Egyptians and Japanese,''
  \textit{PALADYN Journal of Behavioral Robotics},
  vol. 4,
  no. 2,
  pp. 83--93,
  2013

\bibitem{Watanabe15}
  M. Watanabe, K. Ogawa, and H. Ishiguro,
  ``Can androids be salespeople in the real world?,''
  \emph{in Proc. Annual ACM Conference on Human Factors in Computing Systems},
  pp. 781--788,
  2015

\bibitem{Tonkin17}
  M. Tonkin \emph{et al.},
  ``Would you like to sample? robot engagement in a shopping centre,''
  \textit{in Proc. IEEE International Symposium on Robot and Human Interactive Communication},
  pp. 42--49,
  2017

\bibitem{Soderlund20}
  M. S\"{o}derlund,
  ``Employee encouragement of self-disclosure in the service encounter and its impact on customer satisfaction'',
  \textit{Journal of Retailing and Consumer Services},
  vol. 53,
  article 102001,
  2020

\bibitem{Kim21}
  S. S. Kim, J. Kim, F. B.-Baiden, M. Giroux, and Y. Choi,
  ``Preference for robot service or human service in hotels? Impacts of the COVID-19 pandemic'',
  \textit{International Journal of Hospitality Management},
  vol. 93,
  article 102795,
  2021

\bibitem{NUITRACK}
  NUITRACK,
  \url{https://nuitrack.com/}

\bibitem{Admoni17}
  H. Admoni, B. Scassellati,
  ``Social eye gaze in human-robot interaction: A review,''
  \emph{Journal of Human-Robot Interaction},
  vol. 6,
  no. 1,
  pp. 25--63,
  2017

\bibitem{youtube}
  ACM SIGCHI,
  \textit{Please Listen to Me: How to Make Passersby Stop by a Humanoid Robot in a Shopping Mall},
  (Apr. 15, 2020),
  Accesed: Jan. 8, 2021,
  [Online Video],
  Available: https://youtu.be/k7OQa5ux8e4
  
\bibitem{Saad19}
  E. Saad, M. A. Neerincx, K. V. Hindriks,
  ``Welcoming robot behaviors for drawing attention,''
  \textit{In ACM/IEEE International Conference on Human-Robot Interaction},
  pp. 636--637,
  2019
  
\bibitem{Leite08}
  I. Leite, A. Pereira, C. Martinho, and A. Paiva ,
  ``Are Emotional Robots More Fun to Play With?,''
  \emph{in Proc. IEEE International Symposium on Robot and Human Interactive Communication},
  pp. 77--82,
  2008
  
\bibitem{Khaoula14}
  Y. Khaoula, N. Oshima, P. Ravindra, S. De Silva, and M. Okada,
  ``Concepts and Applications of Human-Dependent Robots,'' in
  \emph{Human Interface and the Management of Information. Information and Knowledge in Applications and Services},
  S. Yamamoto, Eds,
  Springer,
  vol. 8522
  pp. 435--444,
  2014

\bibitem{Michalowski07}
  M. P. Michalowski, S. Sabanovic, and H. Kozima,
  ``A Dancing Robot for Rhythmic Social Interaction,''
  \emph{in Proc. International Conference on Human-Robot Interaction},
  pp. 89--96,
  2007
  
\bibitem{Mutlu06}
  B. Mutlu, J. Forlizzi, J. Hodgins,
  ``A storytelling robot: Modeling and evaluation of human-like gaze behavior'',
  \textit{In 2006 6th IEEE-RAS International Conference on Humanoid Robots},
  pp. 518--523,
  2006

\bibitem{Nomura17}
  T. Nomura,
  ``Robots and gender'',
  \textit{Gender and Genome},
  pp. 18--26,
  2017
  
\bibitem{Okafuji20b}
  Y. Okafuji \emph{et al.},
  ``Can a Humanoid Robot Continue to Draw Attention in an Office Environment?,''
  \textit{Advanced Robotics},
  vol. 34,
  no. 14,
  pp. 931--946,
  2020
  
\bibitem{Gockley05}
  R. Gockley \emph{et al.},
  ``Designing robots for long-term social interaction'',
  \textit{in Proc. IEEE/RSJ International Conference on Intelligent Robots and Systems},
  pp. 2199-2204,
  2005

\bibitem{Salem15}
  M. Salem, G. Lakatos, F. Amirabdollahian and K. Dautenhahn,
  ``Would You Trust a (Faulty) Robot? Effects of Error, Task Type and Personality on Human-Robot Cooperation and Trust'',
  \textit{in Proc. International Conference on Human-Robot Interaction},
  pp. 1-8,
  2015
  
\end{thebibliography}

%
% paper title
% Titles are generally capitalized except for words such as a, an, and, as,
% at, but, by, for, in, nor, of, on, or, the, to and up, which are usually
% not capitalized unless they are the first or last word of the title.
% Linebreaks \\ can be used within to get better formatting as desired.
% Do not put math or special symbols in the title.
\title{Behavioral assessment of a humanoid robot when attracting pedestrians in a mall}

%
%
% author names and IEEE memberships
% note positions of commas and nonbreaking spaces ( ~ ) LaTeX will not break
% a structure at a ~ so this keeps an author's name from being broken across
% two lines.
% use \thanks{} to gain access to the first footnote area
% a separate \thanks must be used for each paragraph as LaTeX2e's \thanks
% was not built to handle multiple paragraphs
%

\author
    {Yuki~Okafuji,
      Yasunori~Ozaki,
      Jun~Baba,
      Junya~Nakanishi,
      Kohei~Ogawa,
      Yuichiro~Yoshikawa,
      Hiroshi~Ishiguro% <-this % stops a space
      \thanks{Y. Okafuji is with the School of Information Science and Engineering, Ritsumeikan University, Shiga, Japan e-mail: yokafuji@fc.ritsumei.ac.jp.}% <-this % stops a space
      \thanks{Y. Ozaki and J. Baba are with AI Lab, CyberAgent, Inc., Tokyo, Japan.}% <-this % stops a space
      \thanks{J. Nakanishi, Y. Yoshikawa, and H. Ishiguro are with the Graduation School of Engineering Science, Osaka University, Osaka, Japan.}
      \thanks{K. Ogawa is with the Graduation School of Engineering, Nagoya University, Aichi, Japan.}
    }
    
% note the % following the last \IEEEmembership and also \thanks - 
% these prevent an unwanted space from occurring between the last author name
% and the end of the author line. i.e., if you had this:
% 
% \author{....lastname \thanks{...} \thanks{...} }
%                     ^------------^------------^----Do not want these spaces!
%
% a space would be appended to the last name and could cause every name on that
% line to be shifted left slightly. This is one of those "LaTeX things". For
% instance, "\textbf{A} \textbf{B}" will typeset as "A B" not "AB". To get
% "AB" then you have to do: "\textbf{A}\textbf{B}"
% \thanks is no different in this regard, so shield the last } of each \thanks
% that ends a line with a % and do not let a space in before the next \thanks.
% Spaces after \IEEEmembership other than the last one are OK (and needed) as
% you are supposed to have spaces between the names. For what it is worth,
% this is a minor point as most people would not even notice if the said evil
% space somehow managed to creep in.



% The paper headers
%\markboth{Journal of \LaTeX\ Class Files,~Vol.~14, No.~8, August~2015}%
%{Shell \MakeLowercase{\textit{et al.}}: Bare Demo of IEEEtran.cls for IEEE Journals}
% The only time the second header will appear is for the odd numbered pages
% after the title page when using the twoside option.
% 
% *** Note that you probably will NOT want to include the author's ***
% *** name in the headers of peer review papers.                   ***
% You can use \ifCLASSOPTIONpeerreview for conditional compilation here if
% you desire.




% If you want to put a publisher's ID mark on the page you can do it like
% this:
%\IEEEpubid{0000--0000/00\$00.00~\copyright~2015 IEEE}
% Remember, if you use this you must call \IEEEpubidadjcol in the second
% column for its text to clear the IEEEpubid mark.



% use for special paper notices
%\IEEEspecialpapernotice{(Invited Paper)}




% make the title area
\maketitle

% As a general rule, do not put math, special symbols or citations
% in the abstract or keywords.
\begin{abstract}
Research currently being conducted on the use of robots as human labor support technology. In particular, the service industry needs to allocate more manpower, and it will be important for robots to support people. This study focuses on using a humanoid robot as a social service robot to convey information in a shopping mall, and the robot's behavioral concepts were analyzed. In order to convey the information, two processes must occur. Pedestrians must stop in front of the robot, and the robot must continue the engagement with them. For the purpose of this study, three types of autonomous behavioral concepts of the robot for the general use were analyzed and compared in these processes in the experiment: active, passive-negative, and passive-positive concepts. After interactions were attempted with 65,000+ pedestrians, this study revealed that the passive-negative concept can make pedestrians stop more and stay longer. In order to evaluate the effectiveness of the robot in a real environment, the comparative results between three behaviors and human advertisers revealed that (1) the results of the active and passive-positive concepts of the robot are comparable to those of the humans, and (2) the performance of the passive-negative concept is higher than that of all participants. These findings demonstrate that the performance of robots is comparable to that of humans in providing information tasks in a limited environment; therefore, it is expected that service robots as a labor support technology will be able to perform well in the real world.
\end{abstract}

% Note that keywords are not normally used for peerreview papers.
\begin{IEEEkeywords}
Social service robot, Advertisement, Drawing attention, Field trial
\end{IEEEkeywords}


\section{Introduction}
\label{sec1}
\IEEEPARstart{S}{ocial} robots have been developed and are presently being used in our daily lives. These social robots are deployed in the service industry for various purposes. For instance, social service robots are used as museum guides~\cite{Shiomi07}, travel guides~\cite{Triebel16}, shopping guides~\cite{Gross08}, and for hotel services~\cite{Osawa17}. Currently, significant advancements in machine learning have improved the performance of robots; however, there are limitations for the tasks that robots can perform. In the long-term perspective, however, robots will gradually improve in performance and become more necessary in our lives. As robots become more widespread in our lives, they are expected to become a labor support technology in society~\cite{Yamazaki12} and provide a new type of customer service as avatar robots~\cite{Baba20}.

Among the various tasks in the service industry, tasks of providing information and presenting advertisements in commercial facilities are expected to be one of the roles that robots can play~\cite{Kanda10}. It is expected that a robot is able to approach a target customer more efficiently by using its embodiment, as opposed to the conventional method of providing information through papers or digital signages. For example, some applications of robots wherein they are used for a variety of services (such as customers favor purchasing status goods and order and eat more food) have been validated~\cite{Mende19}. Moreover, stakeholders, which includes customers, shop managers, and mall managers, are positive about introducing robots in shopping malls~\cite{Niemela17, Niemela19}. Consequently, introducing robots into commercial facilities is not limited to advertising and providing information, but they are also highly expected to be a part of our future. Despite these expectations, it has been reported that social robots in the real world tend to be neglected by users even if the robots talk to them, owing to the limitations of their abilities~\cite{Lee12, Tanaka16}. To maximize the effectiveness of robots, it is necessary to not only improve their capabilities but also to identify the types of behavior they should display. 
%
\begin{figure}[!t]
  \centering
  \includegraphics[width=2.5in]{fig1-eps-converted-to.pdf}
  \caption{One of the experimental scenes in a shopping mall. The autonomous robot attempts to convey information about the mall to pedestrians.}
  \label{fig:scene}
\end{figure}

In this study, therefore, we investigated robot behaviors in terms of providing information and advertising to pedestrians in a shopping mall. The setup of the experiment is presented in Fig~\ref{fig:scene}. In particular, the detailed aims of this study are twofold:
%
\begin{enumerate}
  \item Explore versatile robot behaviors that do not depend on providing specific types of information or advertisements.
  \item Propose and verify the behavioral concepts of the robot rather than detailed behavioral verifications, such as eye-contact and talk timing.
\end{enumerate}
%
We believe that these aims will lead to a discussion on how robot behaviors can be applied in general.

In this study, there are three main steps in which robots successfully provide information. These steps are drawing the attention of pedestrians to the robot, making pedestrians stop in front of the robot, and continuing the engagement until the message is delivered. In these tasks, we believe that getting pedestrians to stop and conveying the information to the end are the most difficult tasks. By providing an opportunity for the robot to perform well for these two tasks, we proposed three types of behavioral concepts. These concepts are active, passive-negative, and passive-positive concepts, and they were verified through a field experiment. Based on these results, we discuss what kind of robot behavior concepts can trigger people to want to communicate with robots. In Section~\ref{sec3}, we perform experiments on the robots of the three types of robot behavior for providing information that have been verified in a shopping mall. We also examined whether the proposed robot system was more effective than human advertisers in terms of providing information. For robots to support human labor, they should to demonstrate performance equivalent to that of humans. In particular, tasks in which the robot has a superior performance must be identified. Therefore, four human advertisers were gathered, and the experiment was conducted under the same conditions as that of the robot. Then, we compared the robot's performance to humans. The human experiments are described in Section~\ref{sec4}. The rest of the papr is structured as follows. In Section~\ref{sec2}, related works are described. In Section~\ref{sec5}, the discussion based on the results in the two experiments is provided. Finally, Section~\ref{sec6} provides the conclusions and describes future work.

The preliminary research for this study was presented at a conference and published in the proceedings~\cite{Okafuji20} in which we reported our limited results of the pedestrians affected by the robot behaviors, shown in Table~\ref{tab:results1} and Fig.~\ref{fig:whole_robot} in Section~\ref{sec3}. The current paper provides a detailed analysis depending on the gender, age, and characteristics of the pedestrians in Section~\ref{sec3}. In addition, the current paper provides a comparison of the performance results between the robots and the humans in Section~\ref{sec4}. The comparative results in Section~\ref{sec4} are crucial for the discussion section as an insight into the competence of robots and the diffusion of robots in society. Based on the added results, the introduction, related works, discussion, and conclusion sections are also refined.


\section{Related Works}
\label{sec2}
\subsection{Service Robot in Real-World Environment}
There are many examples of robots being used to provide information and display advertisements in stores. One of the aims of this research in real-world environments is to build the robot system itself. For instance, experiments on a semi-autonomous robot are supported by human operators for dialogues. These robots have been placed at shopping malls and stations~\cite{Kanda10, Shiomi08}, and a multiple robot system has been implemented in a shopping mall~\cite{Shiomi09}. These studies aimed to verify the effectiveness of the system itself, i.e., whether the robot system can be used in the real-world environment, rather than to verify the effects of the robot's detailed behavior.

Several previous studies focused on comparing and verifying the detailed behaviors of the robots that are effectively used by users in real environments. An example of a robot using its mobility is a robot handing out flyers~\cite{Shi13, Shi18}. The behavior of humans distributing flyers was analyzed, and the analyzed optimal behavior was implemented into a robot and then tested in a shopping mall. The robots in the study applied their mobility capabilities to the maximum effect in the task of distributing flyers.

In contrast, stationary robots (or robots that rarely move) have difficulty in terms of providing information because their actions are limited because they cannot approach pedestrians themselves. These robots need to attract pedestrians and draw them near the robot through their presence and actions. After engaging with pedestrians, a study in a museum~\cite{Yamazaki08} explored the timing of the robot's head and gaze action to increase a pedestrian's engagement. In the shopping mall, robot behaviors that are natural for humans have also been validated when humans approach robots~\cite{Bergstrom09}. With respect to the behavior of the stationary robot before engaging pedestrians, the social presence of robots is important; thus, the talking behavior of a robot has been shown to be effective~\cite{Iwasaki18}. In addition, a looking-back behavior is also effective for gaining pedestrians' attention~\cite{Iwasaki19}.

A more detailed analysis shows that there are five elements of robot behaviors that are important in terms of engaging with users: eye contact, duration of eye contact, distance to users, approaching users, and laughing. Shiomi et al. investigated different robot sizes and different conversational schemes for providing information to examine detailed robot behavior~\cite{Shiomi13}. This study revealed that smaller robots that provide specific information have a high success rate in terms of delivering information. Hayashi et al. validated the effectiveness of providing information in terms of the number of robots and the conversation methods~\cite{Hayashi07}. The results suggest that a passive-social medium, where two robots talk to each other and they can provide information to pedestrians indirectly, is better than an interactive-social medium with the users. There is also a study of virtual agents that can change their behavior that is based on the spatial characteristics of humans to enable robots to engage with humans more effectively~\cite{Michalowski06}.

Previous studies of stationary robots have often focused on developing robots that can gain the attention or engagement of pedestrians by implementing and comparing detailed robot behaviors. However, although specific behaviors such as looking-back behavior tend to encourage pedestrian engagement, other robots may be unable to implement them due to functional limitations. Therefore, to implement common behaviors to more robots, it is important to propose a behavioral concept such as the passive-social medium. If we can propose an effective behavioral concept, we can build a variety of detailed motion patterns according to the characteristics of each robot that is based on the concept. In other words, the concepts can be used more generically. A few studies have compared and validated these behavioral concepts. However, it is well known that studies on human-robot interaction are highly influenced by cultural differences~\cite{Li10, Trovato13}. Therefore, additional studies across various places should be conducted in the future.

\subsection{Performance Comparison between Robots and Humans}
When developing a service robot to operate in a real environment, most research aims to be able to accomplish tasks such as providing information and distributing flyers. However, if robots are used as a labor support technology, it is important to compare the performance of robots and humans, and a few studies aimed at them.

As one of the studies that directly compared task performance, an android robot as a salesperson attempted to sell goods at a department store~\cite{Watanabe15}. This study shows that the android robot is able to sell goods as well as human selling goods performance. Another study compared a teleoperated robot with a human in the task of distributing food samples~\cite{Tonkin17}. By distributing food samples while the robot passively approaches pedestrians, the robot achieves high performance comparable to humans. These differences in performance have been shown to be influenced by the unique characteristics of the robot. Factors such as the eeriness of the robot can cause discomfort to the consumer, and as a result, compensatory consumer response such as ordering more food is facilitated~\cite{Mende19}. In other words, the high task performance of robots is not only due to their high ability, but also due to a variety of other factors.

There are also some studies that have investigated how users feel when robots (and virtual agent) and humans perform the same task, although they are not directly compared in task performance. A comparison of tasks performed by a virtual agent and a human in service encounters shows no difference in terms of service satisfaction~\cite{Soderlund20}. Until recent years, users have shown that human services are preferred over robot services, but due to the influence of COVID-19, robot services have also been shown to be particularly preferred in recent years~\cite{Kim21}. As in these studies, we can develop more valuable robots by not only showing that robots can accomplish tasks but also by comparing their performance with that of humans.

\section{Experiment I: With an Autonomous Robot}
\label{sec3}
This study aims to investigate whether the humanoid robot can make pedestrians stop and maintain engagement until they have delivered their intended message. To achieve this, three types of behavioral concepts were designed and compared.

For our investigations, we conducted an exploratory field experiment in a large shopping mall \footnote{Asia and Pacific Trade Center Co., Ltd. (ATC), Nankou-kita 2-1-10, Suminoe Area, Osaka-shi, Osaka, Japan} during July--August 2019. The humanoid robot was present for three weekdays and weekends and was available for 6 hours a day. The robot was placed in one of the shopping mall's corridors so that visitors, such as families, couples, and friends, could freely interact with the robot. We announced to all the pedestrians through a notification board that this was an experiment, and a video was being recorded along with the sensor data. This study was conducted on an opt-out basis for unwilling participants who wanted to be removed from the video and sensor data. The opt-out process may have changed the pedestrians' behavior, such as the pedestrians who attempted to interact with the robot, but quit the interaction owing to the notification. However, no one asked to delete the record in the experiment; thus, the effect of opt-out on the experimental results is expected to be minimal.

This experiment was approved by the facility authorities in the shopping mall and the Research Ethics Committee from Ritsumeikan University (Reference number: BKC-HitoI-2019-006).

\subsection{System Configuration}
%
\begin{figure}[!t]
  \centering
  \includegraphics[width=2.5in]{fig2-eps-converted-to.pdf}
  \caption{Interaction system. The robot and printer were installed on the desk, and the five RGB-D sensors were set behind the robot.}
  \label{fig:system}
\end{figure}
%
\begin{figure*}[!t]
  \centering
  \includegraphics[width=6.0in]{fig3-eps-converted-to.pdf}
  \caption{Combined image of the five RealSense cameras giving a total 220$^\circ$ FOV.}
  \label{fig:images}
\end{figure*}

We built an interaction system containing a humanoid social robot, five RGB-D image sensors, and a printer on a table, as shown in Fig.~\ref{fig:system}. The robot ``Sota'' that was developed by Vstone Co. Ltd. was used as the social robot in this experiment. The humanoid robot is approximately 0.3 m tall and has functions such as voice and LED-generated facial expressions. The robot includes arms with two degrees of freedom (DOF), head with three DOF, and body gestures with one DOF. Although the robot was equipped with a RGB camera on its head, we used additional cameras owing to the field of view (FOV) of the camera equipped on the robot, resulting in the limitations for these experiments. In terms of the 3D image sensors, we used five Intel RealSense D415 sensors, which can capture a RGB image (FOV per one camera: 69.4$^\circ$ $\times$ 42.5$^\circ$ $\times$ 77$^\circ$) and depth image (FOV per one camera: 65$^\circ$ $\times$ 40$^\circ$ $\times$ 72$^\circ$); the maximum range of the depth sensor was 10 m. A combined image of five cameras is shown in Fig.~\ref{fig:images}, and the horizontal FOV of the five sensors that were used in this study was 220$^\circ$. In addition, the printer was installed next to the robot to print a voucher for people who finished the interaction with the robot. The voucher could be exchanged for a bottle of water.

To generate robot behaviors according to the behavior of the pedestrians, we used the human detector ``NUITRACK,'' which can estimate the posture of the human in the image~\cite{NUITRACK}. From the results of NUITRACK and the depth image, we calculated the 3D coordinates of all the pedestrians that were observed from the robot's coordination. We used a computer (Intel Core i9-9900K CPU, NVIDIA GeForce RTX 2080 Ti GPU) to obtain the human posture of all pedestrians at a rate of 30 frames per second (fps). The limitation area to estimate the human posture was approximately within 4 m from the sensors.


\subsection{Interaction Design}
The robot had three types of behavioral concepts when pedestrians were near the robot: active, passive-negative, and passive-positive concepts. As mentioned in the Related Works section, this study aims to verify the behavioral concepts for the general use, rather than comparing the detailed differences for the behaviors. While exploring the behavioral concepts, we conducted a pre-experiment wherein we remotely controlled the robot at the university and performed a variety of behaviors to attempt to make pedestrians stop. From the results in the pre-experiment, we determined that there are three main methods that many people attempt during the task: active, passive-negative, and passive-positive.

During the active state, many operators that controlled the robot attempted to directly start a dialogue to the pedestrians such as ``Hello! How are you doing?'' or ``Where are you going to?''. Then, the conversation starts with the pedestrian replying to the robot. In contrast, some operators attempted other ways of dialogue initiated from the pedestrian rather than making the robot initiate a dialogue. For instance, the robot keeps muttering to itself ``I'm in trouble'' or dancing. The robot does not start the dialogue with the pedestrian until the pedestrian starts talking to the robot. It is important for the robot to generate an opportunity that makes pedestrians want to talk to the robot. Instead of starting a conversation when the pedestrian responds to the robot, the robot should make pedestrians feel that they have spoken to the robot. We call this method of starting the interaction from the pedestrian as a passive concept. In addition, we differentiated between the negative and positive expressions in the passive method. For instance, ``I'm in trouble'' and dancing were considered to be negative and positive expressions, respectively; thus, these are referred to as passive-negative and passive-positive concepts. In the pre-experiment, both passive behaviors demonstrated good results to attract pedestrians.

In summary, the behavioral concepts proposed in this study are as follows:
\begin{itemize}
  \item Active concept: A state in which a conversation starts when the pedestrian responds to the robot's talk.
  \item Passive-negative concept: A state in which the pedestrian observes the negative state of the robot and a conversation starts when the pedestrian first talks to the robot.
  \item Passive-positive concept: A state in which the pedestrian observes the positive state of the robot and a conversation starts when the pedestrian first talks to the robot.
\end{itemize}

This experiment here aimed to compare three behavioral concepts as the manipulative factors, rather than the differences in the detailed movements of the robots. We verified the influence of the behavioral concepts of the robot by measuring the behavior between participants at the same experimental location on different dates.

\subsection{Procedure}
According to the three proposed concepts that are based on the results of the pre-experiment, we specifically designed the ``greeting behavior'' as active, ``troubling behavior'' as passive-negative, and ``dancing behavior'' as passive-positive. The common robot motion for the three behaviors is face-to-face contact with the pedestrian that is closest to the robot so as to clarify whom the robot targets. Face-to-face contact has proven to be effective in human--robot interaction from various aspects, such as conveying the robot's attention~\cite{Admoni17}. For the greeting behavior, the robot makes hand raising gestures to the pedestrians and says ``Hello! Please talk with me!'' During the troubling state, the robot behaves as if it has a headache and keeps muttering to itself for seeking someone's help ``I'm in trouble. What should I do?'' This creates an opportunity for the pedestrian to want to talk to the robot by pretending that the robot is in trouble. During the dancing mode, the robot dances while singing. This also creates an opportunity for the pedestrian to want to talk to the robot for fun.

These behaviors were performed while the posture for at least one pedestrian was measured (maximum range to obtain the postures was 4 $\times$ 4 m as the lateral and depth direction) until the pedestrian stopped in front of the robot. The robot determined if the pedestrian stopped by examining whether or not the pedestrian stayed in the area of 1.2 $\times$ 2.5 m (lateral and depth direction) from the sensors for 3 s. After the pedestrian stopped in front of the robot, the robot started talking about a store in the shopping mall for 13, 19, or 26 s, depending on the scenario. This talk scenario was generated randomly regardless of the three types of behaviors. While the robot was talking, the interaction system with the passersby was not performed, which is called a passive medium~\cite{Hayashi07}. When the pedestrian finished listening to the robot, the printer next to the robot printed a voucher. The pedestrian could exchange the voucher for a bottle of water in the store that the robot was advertising.

These series of robot behaviors were all performed automatically. As explained above, the robot behavior is not determined by verbal interaction with the pedestrian. This is because the environment of the commercial facility was noisy, and the robot system could not accurately recognize what the pedestrians talk to the robot. Therefore, the robot behavior was automatically generated by estimating the pedestrian's state based only on the posture data. Each behavior in motion can be watched from a video that was presented in the preliminary study~\cite{youtube}.

\subsection{Hypothesis}
The greeting behavior was considered a basis because this is often used to get the attention of pedestrians (e.g.,~\cite{Saad19}). On the other hand, previous studies have demonstrated that emotional robots~\cite{Leite08} and human-dependent robots~\cite{Khaoula14}, which are similar to the idea of troubling (passive-negative), can have a higher engagement with users. Therefore, we can expect that troubling also demonstrates a higher performance in this experiment. In addition, the high performance by the dancing behavior (passive-active) is also expected because dancing for rhythmic interaction was performed with higher engagement with children~\cite{Michalowski07}.
%
\begin{figure}[!t]
  \centering
  \includegraphics[width=3in]{fig4-eps-converted-to.pdf}
  \caption{Paths of all the labels that were annotated from the recorded video. The red and blue paths represent the stop rate (SR) and the distribution success rate (DSR). The path through red-yellow-blue denotes the whole distribution success rate (WDSR).}
  \label{fig:label}
\end{figure}
%
\begin{table*}[!t]
  \renewcommand{\arraystretch}{1.3}
  \caption{Total number of pedestrians that walked in front of the robot, pedestrians who stopped in front of the robot, pedestrians stopped while excluding only children, and pedestrians who received the voucher}
  \label{tab:results1}
  \centering
  \begin{tabular}{llcccc}
    \hline
    \multicolumn{2}{c}{Conditions} & Whole Pedestrians & Pedestrians Stopped & Pedestrians Stopped excluding only children & Pedestrians Received \\ \hline 
    \multirow{2}{*}{Greeting}  & Weekday & 6306  & 362  & 273 & 94  \\
                               & Weekend & 19738 & 1139 & 925 & 312 \\
    \multirow{2}{*}{Troubling} & Weekday & 6000  & 559  & 443 & 186 \\
                               & Weekend & 10205 & 1162 & 953 & 396 \\
    \multirow{2}{*}{Dancing}   & Weekday & 7509  & 819  & 644 & 169 \\
                               & Weekend & 16884 & 1057 & 819 & 211 \\
    \hline
  \end{tabular}
\end{table*}

\subsection{Measurement}
Throughout the experiment, we recorded videos as shown in Fig.~\ref{fig:images} to analyze the behavior of all the pedestrians. Thereafter, we labeled all the pedestrians that walked in front of the robot front, the pedestrians that stopped in front of the robot, and the pedestrians who received the voucher. Along with counting the people that stopped in front of the robot, we labeled pedestrians that stopped while excluding situations where there were only children. The reason for excluding only children from the ``pedestrian stopped'' situation was that the human detector NUITRACK cannot detect small children due to the occlusion by the robot and the desk. Thus, the proposed robot system was not able to proceed to the information provision phase even if the children were in front of the robot. The number of people received was labeled depending on the talk scenarios (A: 13 s, B: 19 s, and C: 26 s). The detailed path of all labels is shown in Fig.~\ref{fig:label}.

The features extracted from the recorded videos include the labeled behavior, apparent gender of the pedestrian(s), and estimated age (child under 12 years old or an adult). This annotation was applied to all the pedestrians who passed by the robot in the video. If the same person appeared more than once a day, each appearance was annotated without personal identification. To ensure valid results, video data annotation was performed by two coders. One was the author, Y. O., and the other was a person unrelated to this study, who was hired as a part-time worker. In order to ensure uniformity in the criteria while judging ``stop'' behaviors in front of the robot, the two coders worked together to determine the criteria before annotating. The data for one day were overlapped, and the analysis of the overlapped data showed that they were well matched (Cohen's Kappa was .894).

\begin{figure*}[!t]
  \centering
  \subfloat[SR]{\includegraphics[width=2.3in]{fig5a-eps-converted-to.pdf}
    \label{fig:SR_robot}}
  \hfil
  \subfloat[DSR]{\includegraphics[width=2.3in]{fig5b-eps-converted-to.pdf}
    \label{fig:DSR_robot}}
  \hfil
  \subfloat[WDSR]{\includegraphics[width=2.3in]{fig5c-eps-converted-to.pdf}
    \label{fig:WDSR_robot}}
  \caption{Results of the stop rate (SR), distribution success rate (DSR), and the whole distribution success rate (WDSR) according to each robot behavior in Experiment I.}
  \label{fig:whole_robot}
\end{figure*}

We used three indexes for the evaluation: the stop rate (SR), distribution success rate (DSR), and whole distribution success rate (WDSR). The SR is the ratio of the number of pedestrians that stopped to the number of all pedestrians (red path in Fig.~\ref{fig:label}). The DSR is the ratio of the number of pedestrians who received the voucher to the pedestrians that stopped, while excluding only children (blue path in Fig.~\ref{fig:label}). The WDSR is the ratio of the number of pedestrians who received the voucher to the number of all pedestrians (path through red-yellow-blue in Fig.~\ref{fig:label}).

\subsection{Results}
\begin{figure*}[!t]
  \centering
  \subfloat[SR]{\includegraphics[width=2.3in]{fig6a-eps-converted-to.pdf}
    \label{fig:gender_SR_robot}}
  \hfil
  \subfloat[DSR]{\includegraphics[width=2.3in]{fig6b-eps-converted-to.pdf}
    \label{fig:gender_DSR_robot}}
  \caption{Results of the stop rate (SR) and the distribution success rate (DSR) according to the gender in Experiment I.}
  \label{fig:gender_robot}
\end{figure*}

\begin{figure*}[!t]
  \centering
  \subfloat[SR]{\includegraphics[width=2.3in]{fig7a-eps-converted-to.pdf}
    \label{fig:age_SR_robot}}
  \hfil
  \subfloat[DSR]{\includegraphics[width=2.3in]{fig7b-eps-converted-to.pdf}
    \label{fig:age_DSR_robot}}
  \caption{Results of the stop rate (SR) and the distribution success rate (DSR) according to the age in Experiment I.}
  \label{fig:age_robot}
\end{figure*}

\begin{figure}[!t]
  \centering
  \includegraphics[width=2.5in]{fig8-eps-converted-to.pdf}
  \caption{Results of the distribution success rate (DSR) according to each scenario in Experiment I.}
  \label{fig:scenario_robot}
\end{figure}

\begin{figure*}[!t]
  \centering
  \subfloat[SR]{\includegraphics[width=2.4in]{fig9a-eps-converted-to.pdf}
    \label{fig:time_SR_robot}}
  \hfil
  \subfloat[DSR]{\includegraphics[width=2.4in]{fig9b-eps-converted-to.pdf}
    \label{fig:time_DSR_robot}}
  \caption{Results of the stop rate (SR) and the distribution success rate (DSR) according to the time window in Experiment I.}
  \label{fig:time_robot}
\end{figure*}

\begin{table*}[!t]
  \renewcommand{\arraystretch}{1.3}
  \caption{Ratio of the number of times between one pedestrian and the group that stopped in front of the robot and listened to the robot's message until the end.}
  \label{tab:results2}
  \centering
  \begin{tabular}{llcccc}
    \hline
    \multicolumn{2}{c}{Conditions} & \multicolumn{2}{c}{Ratio of pedestrians stopped} & \multicolumn{2}{c}{Ratio of pedestrians received} \\
    & & One pedestrian \% & Two or more pedestrians \% & One pedestrian \% & Two or more pedestrians \% \\ \hline
    \multirow{2}{*}{Greeting}  & Weekday & 36.2 & 63.8 & 13.5 & 86.5 \\
                               & Weekend & 35.1 & 64.9 & 11.0 & 89.0 \\
    \multirow{2}{*}{Troubling} & Weekday & 39.5 & 60.5 & 10.0 & 90.0 \\
                               & Weekend & 34.3 & 65.7 & 8.3  & 91.7 \\
    \multirow{2}{*}{Dancing}   & Weekday & 37.3 & 62.7 & 12.5 & 87.5 \\
                               & Weekend & 38.1 & 61.9 & 10.0 & 90.0 \\
    \hline
  \end{tabular}
\end{table*}

The results for the number of labeled pedestrians are shown in Table~\ref{tab:results1}. The results for the SR, DSR, and WDSR according to each robot behavior are shown in Fig.~\ref{fig:whole_robot}. We verified the differences in the number of pedestrians among the behavior conditions with a Chi-squared test. We used the Cramer's $V$ as the effect size in all the Chi-square test results. The results revealed significant differences among the behavioral conditions: $(\chi^2(2) = 333.64, p < 0.01, V = 0.05)$ in the SR, $(\chi^2(2) = 81.61, p < 0.01, V = 0.10)$ in the DSR, and $(\chi^2(2) = 252.14, p < 0.01, V = 0.04)$ in the WDSR. The residual analysis in comparison with the mean across all behaviors showed that (1) the greeting had low SR and WDSR ratios, (2) the troubling state had high SR, DSR, and WDSR ratios, and (3) dancing had a high SR ratio but low DSR and WDSR ratios. Therefore, our results show that the robot motion that behaves as though it is in trouble makes pedestrians stop more and stay longer in front of the robot in comparison to the greeting and dancing behaviors. Meanwhile, the dancing behavior resulted in a significant number of pedestrians to stop, but the stopped pedestrians did not listen to the robot talk for a long time. 

As described in the detailed analysis, the additional results of the gender difference in the SR and DSR, age difference in the SR and DSR, and scenario difference in the DSR are shown in Figs.~\ref{fig:gender_robot}-\ref{fig:scenario_robot}. We also verified the differences in the number of people for the gender in the SR and DSR, age in the SR and DSR, and scenario in DSR through a Chi-square test. The results among the gender revealed significant differences in the SR $(\chi^2(1) = 21.50, p < 0.01, V = 0.02)$ and the DSR $(\chi^2(1) = 5.26, p = 0.02, V = 0.04)$. The results among the age showed significant differences in the SR $(\chi^2(1) = 2666.67, p < 0.01, V = 0.20)$, but no significant differences in the DSR $(\chi^2(1) = 1.10, p = 0.29, V = 0.02)$. As a result, this indicates that females and children are more likely to stop in front of the robot. However, there were no gender or age differences in whether they listened to the message to the end. In terms of the talk scenario difference (Scenario A: 13 s, Scenario B: 19 s, Scenario C: 26 s), there was a significant difference in the DSR $(\chi^2(2) = 77.37, p < 0.01, V = 0.12)$. In addition, the findings revealed that the pedestrians listen to the robot talk in its entirety as the scenario becomes shorter.

Next, there were differences in pedestrian behavior in the hourly results; Fig.~\ref{fig:time_robot} shows the SR and DSR results for each hour. As the start time of the experiment was different for each condition, we validated the results from 12--4 pm, which is the common time window when the experiment was conducted, with a Chi-square test. The results among the time revealed significant differences in the SR $(\chi^2(4) = 65.76, p < 0.01, V = 0.02)$ and the DSR $(\chi^2(4) = 21.28, p < 0.01 , V = 0.04)$. The residual analysis in comparison with the mean across all times showed that (1) 12 pm had low SR ratio, (2) 1 pm had low DSR ratio, and (3) 3 pm and 4 pm had high SR and DSR ratios. Therefore, these results indicate that it is difficult for robots to approach pedestrians during the hours close to lunchtime.

Finally, we show the ratio of the number of times between one pedestrian and the group that stopped in front of the robot and listened to its message until the end, as shown in Table~\ref{tab:results2}. The results revealed that the differences are small for the behavioral conditions in terms of the pedestrians stopping rates and the received pedestrians. In addition, it was determined that pedestrians in a group stop in front of the robot and listen to its entire message in comparison to those who are alone.


\section{Experiment II: By Humans}
\label{sec4}
Experiment I aims to identify robot behaviors that can make pedestrians stop and maintain engagement with them. The results show that the troubling behavior can attract more pedestrians based on a high SR, DSR, and WDSR. Meanwhile, the results of the troubling behavior are based on the relative results in comparison to the two other types of robot behaviors. Therefore, we cannot argue that these robot behaviors are more competent than human behaviors. If robots are to be used as a labor support technology, a better performance by robots then humans is desirable. In addition, clarifying the tasks wherein robots are superior to humans provides invaluable insights in collaborating with humans. Several studies have been conducted to examine whether robots can play an active role in the real world by comparing the performance of robots and humans (e.g.~\cite{Watanabe15}). Thus, to compare the results that are generated by the robots and those from humans, we conducted a second experiment in which the robot was replaced by a person in an equivalent experimental environment.

The experiment was conducted in the same location as Experiment I in November 2019. With the same situation as Experiment I, the experiment was recorded with a notification board and conducted on an opt-out basis.

This experiment was approved by the facility authorities in the shopping mall and the Research Ethics Committee from Ritsumeikan University (Reference number: BKC-HitoI-2019-006-1).


\subsection{Human Advertisers}
Four people who have experience in distributing flyers were recruited through a temporary agency to participate in the experiment (two males/females, average age: 23.75 years old). The experiment was conducted over 4 days, and each human advertiser performed the task each day. All the human advertisers provided informed consent, which allowed for the use of the collected data for scientific purposes and publication. The human advertisers received 9,500 JPY a day, and they also received a reward in accordance to their performance.


\subsection{Interaction Design}
The human advertisers were instructed to encourage the pedestrians to stop to provide information about the shop. The advertisers were allowed to use the leaflet of the shop, which is different from Experiment I. This task requires the human advertisers to perform their usual way of providing information. As one of the goals in Experiment II was to compare the robot's performance with the human's usual performance, human behavior should be as unrestricted as possible. After the pedestrians were stopped by the advertiser's attempt, the advertiser must provide them with information about the shop, such as recommended products. It was forbidden to annoy the pedestrians to force them to stop, such as interfering with their walking or saying that water is being distributed. When the pedestrian finished listening to the information about the shop, the advertiser gave a voucher that could be exchanged for a bottle of water. 

As a standard for the area in which the advertiser could move during the experiment, an area of 1.2 $\times$ 2.5 m (lateral and depth direction) was specified, which is the same area for determining the pedestrians to stop in Experiment I. Thus, the significant difference between Experiments I and II was whether to use the leaflet, and other situations were set up to be similar. However, we treated Experiments I and II as separate experiments because the conditions were not completely consistent.

The human advertisers were allowed to practice the experimental task for approximately 20 min before the experiment. The advertisers performed three sets of 50 min of executing the experimental task followed by 10 min of rest. To provide motivation to the advertisers, an additional reward of 500 JPY for every 10 vouchers were given to the advertisers. The scene during Experiment II is shown in Fig.~\ref{fig:Ex2}.
%
\begin{figure}[!t]
  \centering
  \includegraphics[width=2.5in]{fig10-eps-converted-to.pdf}
  \caption{Example scene in Experiment II with a human advertiser conveying information to pedestrians.}
  \label{fig:Ex2}
\end{figure}

\subsection{Measurement}
We labeled some indices from the recorded video, which are similar to the labels in Experiment I, as shown in Fig.~\ref{fig:label}. While the robot system in Experiment I could not recognize the situation with only small children, ``Pedestrians stopped excluding only children'' was not labeled because that situation does not occur in Experiment II. The time during which a pedestrian stops in front of the advertiser was also annotated, instead of the paths of each scenario. This annotation was applied to all the pedestrians who passed in front of the robot in the video. To ensure valid results, video data annotation was also performed by two coders. One was the author, Y. O., and the other was a person unrelated to this study. The data for 1 day overlapped, and the analysis of the overlapped data indicates that they were well matched (Cohen's Kappa was .917).


\subsection{Results}
%
\begin{table}[!t]
  \renewcommand{\arraystretch}{1.3}
  \caption{Total number of pedestrians, pedestrians who stopped, and pedestrians who received the voucher.}
  \label{tab:results3}
  \centering
  \begin{tabular}{lcccc}
    \hline
     & Whole & Pedestrians & Pedestrians \\
    Human Advertisers & Pedestrians & Stopped & Received \\ \hline 
    1 (Weekday) & 1312 & 46  & 28  \\
    2 (Weekday) & 1630 & 57  & 25  \\
    3 (Weekend) & 8582 & 70  & 36  \\
    4 (Weekend) & 7103 & 290 & 205  \\
    \hline
  \end{tabular}
\end{table}
%
\begin{figure*}[!t]
  \centering
  \subfloat[SR]{\includegraphics[width=2.3in]{fig11a-eps-converted-to.pdf}
    \label{fig:SR_human}}
  \hfil
  \subfloat[DSR]{\includegraphics[width=2.3in]{fig11b-eps-converted-to.pdf}
    \label{fig:DSR_human}}
  \hfil
  \subfloat[WDSR]{\includegraphics[width=2.3in]{fig11c-eps-converted-to.pdf}
    \label{fig:WDSR_human}}
  \caption{Results of the stop rate (SR), distribution success rate (DSR), and the whole distribution success rate (WDSR) according to each human advertiser in Experiment II.}
  \label{fig:whole_human}
\end{figure*}
%
\begin{figure}[!t]
  \centering
  \includegraphics[width=2.5in]{fig12-eps-converted-to.pdf}
  \caption{Results of the duration when successfully making the pedestrians stop during in Experiment II. The error bars represent the standard error of the mean.}
  \label{fig:duration_human}
\end{figure}

The results on the number of labeled pedestrians are shown in Table~\ref{tab:results3}. The results of each SR, DSR, and WDSR according to each advertiser are shown in Fig.~\ref{fig:whole_human}. The results show that there are large individual differences in all the SR, DSR, and WDSR evaluations. Advertiser 3 had the lowest SR, and Advertiser 2 had the lowest DSR. In terms of the WDSR as the total index to provide information, Advertiser 4 was able to hand out the most vouchers to the pedestrians. In an additional analysis, we measured how long the pedestrians listened to the advertiser's message when the pedestrian got the voucher, as shown in Fig.~\ref{fig:duration_human}. Advertiser 4 required the least amount of time to provide information before handing out the vouchers, with an average time of less than 20 s. These results indicate that the shorter the time to provide the information, the better the DSR.

Next, to examine the aims of Experiment II, we compared the results that were generated by the robot behaviors and that of the human advertisers. In terms of the SR, the average results of the robots for the weekday and weekend were 5.7, 10.6, and 7.7 \% for the greeting, troubling, and dancing behaviors, respectively. In contrast, the results by each advertiser were 3.5, 3.5, 0.8, and 4.1 \%, respectively. All the results in the SR by the robot outperform the human performance, and the statistical results with the Chi-squared test with the Cramer's $V$ compared the total values of the robots to the advertisers also displayed significant differences: $(\chi^2(1) = 636.84, p < 0.01, V = 0.08)$. In other words, the robots can perform the stopping task easier.

With the DSR, the average results of the robots for the weekday and weekend were 33.9, 42.0, and 26.0 \% for the greeting, troubling, and dancing behaviors, respectively. In contrast, the results by each advertiser were 60.9, 43.9, 51.4, and 86.2 \%. All the results in the DSR by the advertisers outperform the robot performance. The results of comparing the total values of the robots and the advertisers also show significant differences: $(\chi^2(1) = 158.51, p < 0.01, V = 0.19)$. This is the opposite result of the SR. We also measured the length of the stop time by randomly sampling approximately 20\% of the pedestrians who stopped in front of the robot in Experiment I. The results show that the average length of stop time for all robot behaviors was 21.5 s, whereas the average length of the stop time for all human advertisers was 22.0 s. We verified the differences in these results with a non-paired t-test, and the result showed no significant differences: $(t(678) = 0.25, p = 0.80, d = 0.02)$. This indicates that, although the length of stop time is the same for robots and human advertisers, the DSR is higher for the latter.

With the WDSR as the total index in providing the information, the average results by the robots during the weekday and weekend were 1.6, 3.6, and 1.6 \% for the greeting, troubling, and dancing behaviors, respectively. In contrast, the results by each advertiser were 2.1, 1.5, 0.4, and 3.5 \%. The results of the greeting and dancing behaviors are comparable to the average human performance. On the other hand, in terms of the troubling behavior, the robot has a higher performance than Advertiser 4, who demonstrated the best performance among the advertisers. The results in comparing the total values of the robots and the advertisers indicate significant differences: $(\chi^2(1) = 17.14, p < 0.01, V = 0.01)$. Consequently, these results show that the robots are able to perform similar to humans or even better.

\begin{table*}[!t]
  \renewcommand{\arraystretch}{1.3}
  \caption{Ratio of the number of times between one pedestrian and the group that stopped in front of the human advertiser and listened to the their message.}
  \label{tab:results4}
  \centering
  \begin{tabular}{ccccc}
    \hline
    Human & \multicolumn{2}{c}{Ratio of pedestrians stopped} & \multicolumn{2}{c}{Ratio of pedestrians received} \\
    Advertisers & One pedestrian \% & Two or more pedestrians \% & One pedestrian \% & Two or more pedestrians \% \\ \hline
    1 (Weekday) & 65.6 & 34.4 & 52.9 & 47.1 \\
    2 (Weekday) & 59.5 & 40.5 & 64.7 & 35.3 \\
    3 (Weekend) & 17.1 & 82.9 & 16.7 & 83.3 \\
    4 (Weekend) & 33.6 & 66.4 & 34.1 & 65.9 \\
    \hline
  \end{tabular}
\end{table*}

Finally, we show the ratio of the number of times between one pedestrian and the group that stopped in front of the human advertiser and listened to their message, as shown in Table~\ref{tab:results4}. The results revealed that the differences between the strategies of each human advertiser are large. Advertisers 1 and 2 had a large percentage of successful stops and distributions for one pedestrian because they likely tend to talk to a large number of individual pedestrian. Meanwhile, Advertisers 3 and 4 had a large percentage of stops and successful distributions for groups. These results in Experiment II differed from Experiment I, indicating that the results are independent of the robot behaviors.

\section{Discussions}
\label{sec5}
\subsection{Discussion on Experiment I}
This study mainly aims to investigate whether the humanoid robot can make pedestrians stop in front of it and maintain engagement with them. Thus, we designed three behaviors: greeting as an active concept, troubling as a passive-negative concept, and dancing as a passive-positive concept.

The results in Experiment I show that when the robot exhibited the troubling behavior, it had the best performance among the proposed behaviors in terms of attracting pedestrians and providing information. In addition, the dancing behavior also showed a similar high performance in comparison to troubling only to make people stop. The difference in these results is probably due to the different reasons why the pedestrians approached the robots. When humans see the weak or human-dependent robots, they tend to increase their engagement with the robots because humans want to help robots~\cite{Khaoula14}. This is similar to a phenomenon wherein an adult reaches out to a child when the child is in trouble. Therefore, it seems that many pedestrians were willing to listen to the robot in the troubling situation because the troubling behavior is a similar case. In contrast, in the dancing behavior, we assume that the most common reason that pedestrians approached the robot is for fun. However, despite that they approached the robot to enjoy its dance, the pedestrians felt disconnected when the robot started talking about the shop's information after they approached it. Thus, it can be assumed that the DSR was the lowest among all the proposed behaviors owing to this gap. Consequently, these results suggest that a passive concept (troubling and dancing) is effective to make the pedestrians stop, when compared to the active concept (greeting). In addition, in the passive concept, the consistency in the robot's behavior is crucial in maintaining its engagement with the pedestrians.

Other interesting results are the gender and age differences in the SR and DSR. The results of the SR and DSR showed that women, as well as children, mostly stop in front of the robot and listen when the robot talks. The same situation has also occurred in other studies~\cite{Bergstrom09}, wherein children, sometimes accompanied by their parents, often interact with the robots. In addition, the results of the previous studies indicated that men have a more positive attitude toward interacting with robots than women~\cite{Mutlu06, Nomura17}. However, other studies that use the robot ``Sota,'' which is the same as used in the present study,~\cite{Okafuji20b} showed that women are more interested in the robot; this finding is consistent with our results. Therefore, the results suggest that gender difference in terms of the interest in the robots does not give a definite decision, but it may depend on the appearance of the robot. Meanwhile, the results of the DSR did not show a significant difference by the age. This robot was a non-interactive robot; that is, it did not have the ability to communicate with humans through dialogues. In this case, we assume that the results of the DSR represent the pedestrian's time where they lost interest in the robot and it was similar regardless of the age. In other words, even if the robots are interacting with a person who is strongly interested in the robot, a robot with a poor interactive ability will quickly get boring.

Finally, it was determined that, when pedestrians are in a group, they are more likely to stop and are more willing to listen to the full message. Several examples of these observations have been reported in other studies~\cite{Shi18, Michalowski06, Tonkin17, Okafuji20b}. These observations were thought to be caused by another person who draws the pedestrian to interact with the robot, and not by the robot itself~\cite{Shi18, Michalowski06}. Therefore, in this study, the reason for this observation can be assumed to be that if one person in the group is interested in the robot, the others should listen to the robot; thus, they will wait for him and/or her. Therefore, in crowded situations, such as shopping malls, it is more efficient if the robots talk to the group to provide the information than to an individual person.

\subsection{Discussion on Experiment II}
We also compared the results by the robots and the humans in Experiment II. When comparing both their performance, the SR was higher for the robots and the DSR was higher for the humans. The reason for the low DSR can be assumed to be that the robot's verbal interaction ability was significantly lower than that of humans. It was not possible to implement verbal interaction in the robotic system because the robots do not exhibit a proper speech recognition in a noisy commercial environment. In future studies, this problem can be solved by developing a technology that can recognize the speech of pedestrians correctly even in noisy environments.

In contrast, there are several possible reasons why, from the SR perspective, the robots can perform better than humans. The first is simply because pedestrians were strongly interested in the robot's behavior. The greeting behavior, which is the lowest SR among the robot behaviors, is able to improve the SR in comparison to humans owing to its novelty effects~\cite{Gockley05}. In addition, other behaviors were able to improve the SR drastically. Thus, the different types of robot behaviors can attract pedestrians. The second was that the pedestrians did not prefer the situation wherein the adult's human advertisers were calling out to them to provide information. As mentioned in the discussion on Experiment I, an adult tends to reach out to a child when the child is in trouble. On the other hand, in the case of Japan, where the experiment was conducted, when a strange adult talks to a person in a shopping mall or in a town, the stranger is often a salesperson who recommends something. This is considered nuisance to the pedestrians. Therefore, pedestrians avoid conversations with strangers before they know the type of information will be provided. Accordingly, we believe that the SR of the adult advertisers was lower; however the SR could be higher if a child advertiser did the same task (if it were ethically possible).

These results may be useful in the future collaborative design of robots to support humans. For example, by having a remote avatar robot system such as in~\cite{Baba20, Shiomi08}, it is possible to build a more high-performance system by integrating the capabilities of robots with those of humans. In this study, the environment was a noisy shopping mall in which it is difficult to recognize the speech of a particular individual. Thus, we constructed a passive medium system in which the robot talks unilaterally. However, when it comes to interacting with pedestrians, humans can interpolate the dialogue to improve the interactivity. The autonomous robot attempts to make pedestrians stop in front of the robot, and humans are interpolated in situations where it is difficult for the autonomous robot to talk with the pedestrians. Thus, we expect to build a system that can demonstrate a high performance by interpolating the weaknesses of the robots and humans.

The results of the WDSR for the greeting and dancing behaviors of the robots are comparable to human performance; however, the troubling behavior performance exceeded all the results of the human advertisers. While the SR and DSR are part of the process in evaluating the performance of an informational task, the WDSR is the final evaluation of performance. In other words, in this experiment, the robot succeeded in providing more information to the pedestrians than the human advertisers. Therefore, the results may indicate that robots are more effective in the information provision task.

In addition, this result does not consider the decrease in human performance over time. In this experiment, the advertisers were asked to perform the tasks for 3 h, and performance degradation during this time was not measured. However, if they perform the task over a longer period, performance degradation due to fatigue occurs. In that case, the robots can deliver better results than humans in the work environment. 

In Experiment II, we observed that the SR and WDSR of Advertiser 3 were lower than those of the other advertisers. This is because human advertiser 3 approached pedestrians less often than the other advertisers (Number of people passing per approach for each advertiser: 4.6, 5.6, 46.9, and 10.9, respectively). On the other hand, this phenomenon does not occur with robots; thus, robots have the advantage in that their performance is not affected by factors such as human individual differences. In summary, the results from this study suggest that robots can be sufficient as a labor support technology, which is one of the goals of robotic research.

\subsection{Limitations}
Finally, we want to present the limitations of this study. First, we compared the performances of the three types of proposed behavioral concepts of robots. This is because we considered that the behavioral concepts can be used generically, which would be more effective in comparison to detailed robot behavior. However, it is unclear whether two types of robots that are implemented with the same behavioral concept but have slightly different details of motion can achieve similar results. Even when we use the passive method, we can design other behaviors than dancing. This is a limitation that we need to explore in future studies, which also considers the difference in the robot's appearance and degrees of freedom. 

In addition, this study cannot demonstrate whether the proposed behavioral concept always shows the same results. In this study, we conducted the experiment with the robot in a commercial facility where many people have relatively more time to spare. However, through the experiments, we found that most of the approaches from the robot failed for people in a hurry. Therefore, depending on environmental conditions such as the context and location where the robot is installed, we cannot guarantee if the troubling behavior has as a significant effect on people as the results of this study.

Next, our results strongly depend on the novelty effect. In the troubling behavior, the robot said ``I'm in trouble'' to attract pedestrians' attention, and then they conveyed information about the store. This may have been a type of ``crying wolf.'' Further, in the field of HRI, for example, it has been reported that robot errors decrease people's trust in the robot~\cite{Salem15}. In other words, when people are exposed to the troubling behavior more than once, their trust in the robot may decrease, and they may not listen to the robot. In such a situation, the robot may not be able to surpass human performance even with SR that have shown superior performance over humans. This long-term performance will be considered for future studies.

We should also consider that the results of this study are highly dependent on cultural differences. Previous studies on human-robot interaction with several cultural differences have shown that people with different cultures behave differently depending on the task and the appearance of the robot~\cite{Li10, Trovato13}. For example, for people who think robots are mechanical rather than humans, a robot's troubling behavior may seem creepy. In this case, troubling behavior could deliver the worst result. In human--human interaction, we showed that pedestrians in Japan tend to avoid talking with strangers. However, in other cultures, robots may not be able to outperform humans in terms of the SR results. During this experiment, we did not interview any pedestrians who interacted with the robot. As we cannot infer how the pedestrians felt through their interaction with the robot, we cannot have a rigorous discussion on the cultural differences that affected them. Therefore, the cultural difference is another limitation of this study.

From a perspective closer to cultural differences, the degree to which people are accustomed to robots affects the results of this study. In today's society, social robots are still a rarity and an intriguing object. However, we believe that in a future society where a variety of social robots are prevalent, people care less about what a robot does even if it dances or behaves as it is in trouble. Therefore, we need to consider that, in the future, results may be different from those obtained in this study.


\section{Conclusion}
\label{sec6}
This study investigated whether a humanoid robot can make pedestrians stop in front of it and listen to its message. We proposed three types of behavioral concepts for the robots for the general purpose: active, passive-negative, and passive-positive concepts. According to the three proposed concepts, we specifically designed the ``greeting behavior'' as active, ``troubling behavior'' as passive-negative, and ``dancing behaviors'' as passive-positive. The robot with each behavior was placed in a shopping mall, and the effectiveness of the robots for providing information was verified.

The results from the exploratory field experiments revealed that the troubling behavior, that is, the robot behaves as if it is in trouble, can make pedestrians stop more and stay longer in front of the robot. These results were compared to the results achieved by four humans under the same situation, in which they attempted to make the pedestrians stop to provide information. The comparative results show (1) the performance of the robots were higher than that of the humans in the stop rate (SR), and (2) in the distribution success rate (DSR), the human performance was better than the robots' performance. In particular, in terms of the whole distribution success rate (WDSR), the performance obtained using the greeting and dancing behaviors of the robots are comparable to the human performance. Furthermore, it was determined that the performance of the troubling behavior was higher than those of all the human advertisers who participated in this experiment. These findings demonstrate that the performance of robots is not inferior to that of humans in providing information task. Therefore, it is expected that service robots are able to perform well in the real world. In other words, the results of this study suggest that robots can be sufficient as a labor support technology, which is one of the goals of robotic research.

This study, however, has some limitations because it is difficult for the robots to interact naturally with pedestrians in a noisy environment. This is because automatic dialogue generation is difficult due to the low accuracy of speech recognition for certain pedestrians in noisy environments. These problems are common to all robots that operate in real environments. Hence, ensuring that the robot can recognize the speech content of only the target person in a noisy environment is required. However, by interpolating the weaknesses of the robots and humans, we can build an integrated robot system that can demonstrate high performance. By achieving this, we believe that it is important for robot designs to compensate for the weaknesses of robots and humans in the future.


% For peer review papers, you can put extra information on the cover
% page as needed:
% \ifCLASSOPTIONpeerreview
% \begin{center} \bfseries EDICS Category: 3-BBND \end{center}
% \fi
%
% For peerreview papers, this IEEEtran command inserts a page break and
% creates the second title. It will be ignored for other modes.
\IEEEpeerreviewmaketitle



%\section{Introduction}
% The very first letter is a 2 line initial drop letter followed
% by the rest of the first word in caps.
% 
% form to use if the first word consists of a single letter:
% \IEEEPARstart{A}{demo} file is ....
% 
% form to use if you need the single drop letter followed by
% normal text (unknown if ever used by the IEEE):
% \IEEEPARstart{A}{}demo file is ....
% 
% Some journals put the first two words in caps:
% \IEEEPARstart{T}{his demo} file is ....
% 
% Here we have the typical use of a "T" for an initial drop letter
% and "HIS" in caps to complete the first word.
%\IEEEPARstart{T}{his} demo file is intended to serve as a ``starter file''
%for IEEE journal papers produced under \LaTeX\ using
%IEEEtran.cls version 1.8b and later.
% You must have at least 2 lines in the paragraph with the drop letter
% (should never be an issue)
%I wish you the best of success.

%\hfill mds
 
%\hfill August 26, 2015

%\subsection{Subsection Heading Here}
%Subsection text here.

% needed in second column of first page if using \IEEEpubid
%\IEEEpubidadjcol

%\subsubsection{Subsubsection Heading Here}
%Subsubsection text here.


% An example of a floating figure using the graphicx package.
% Note that \label must occur AFTER (or within) \caption.
% For figures, \caption should occur after the \includegraphics.
% Note that IEEEtran v1.7 and later has special internal code that
% is designed to preserve the operation of \label within \caption
% even when the captionsoff option is in effect. However, because
% of issues like this, it may be the safest practice to put all your
% \label just after \caption rather than within \caption{}.
%
% Reminder: the "draftcls" or "draftclsnofoot", not "draft", class
% option should be used if it is desired that the figures are to be
% displayed while in draft mode.
%
%\begin{figure}[!t]
%\centering
%\includegraphics[width=2.5in]{myfigure}
% where an .eps filename suffix will be assumed under latex, 
% and a .pdf suffix will be assumed for pdflatex; or what has been declared
% via \DeclareGraphicsExtensions.
%\caption{Simulation results for the network.}
%\label{fig_sim}
%\end{figure}

% Note that the IEEE typically puts floats only at the top, even when this
% results in a large percentage of a column being occupied by floats.


% An example of a double column floating figure using two subfigures.
% (The subfig.sty package must be loaded for this to work.)
% The subfigure \label commands are set within each subfloat command,
% and the \label for the overall figure must come after \caption.
% \hfil is used as a separator to get equal spacing.
% Watch out that the combined width of all the subfigures on a 
% line do not exceed the text width or a line break will occur.
%
%\begin{figure*}[!t]
%\centering
%\subfloat[Case I]{\includegraphics[width=2.5in]{box}%
%\label{fig_first_case}}
%\hfil
%\subfloat[Case II]{\includegraphics[width=2.5in]{box}%
%\label{fig_second_case}}
%\caption{Simulation results for the network.}
%\label{fig_sim}
%\end{figure*}
%
% Note that often IEEE papers with subfigures do not employ subfigure
% captions (using the optional argument to \subfloat[]), but instead will
% reference/describe all of them (a), (b), etc., within the main caption.
% Be aware that for subfig.sty to generate the (a), (b), etc., subfigure
% labels, the optional argument to \subfloat must be present. If a
% subcaption is not desired, just leave its contents blank,
% e.g., \subfloat[].


% An example of a floating table. Note that, for IEEE style tables, the
% \caption command should come BEFORE the table and, given that table
% captions serve much like titles, are usually capitalized except for words
% such as a, an, and, as, at, but, by, for, in, nor, of, on, or, the, to
% and up, which are usually not capitalized unless they are the first or
% last word of the caption. Table text will default to \footnotesize as
% the IEEE normally uses this smaller font for tables.
% The \label must come after \caption as always.
%
%\begin{table}[!t]
%% increase table row spacing, adjust to taste
%\renewcommand{\arraystretch}{1.3}
% if using array.sty, it might be a good idea to tweak the value of
% \extrarowheight as needed to properly center the text within the cells
%\caption{An Example of a Table}
%\label{table_example}
%\centering
%% Some packages, such as MDW tools, offer better commands for making tables
%% than the plain LaTeX2e tabular which is used here.
%\begin{tabular}{|c||c|}
%\hline
%One & Two\\
%\hline
%Three & Four\\
%\hline
%\end{tabular}
%\end{table}


% Note that the IEEE does not put floats in the very first column
% - or typically anywhere on the first page for that matter. Also,
% in-text middle ("here") positioning is typically not used, but it
% is allowed and encouraged for Computer Society conferences (but
% not Computer Society journals). Most IEEE journals/conferences use
% top floats exclusively. 
% Note that, LaTeX2e, unlike IEEE journals/conferences, places
% footnotes above bottom floats. This can be corrected via the
% \fnbelowfloat command of the stfloats package.




%\section{Conclusion}
%The conclusion goes here.





% if have a single appendix:
%\appendix[Proof of the Zonklar Equations]
% or
%\appendix  % for no appendix heading
% do not use \section anymore after \appendix, only \section*
% is possibly needed

% use appendices with more than one appendix
% then use \section to start each appendix
% you must declare a \section before using any
% \subsection or using \label (\appendices by itself
% starts a section numbered zero.)
%


%\appendices
%\section{Proof of the First Zonklar Equation}
%Appendix one text goes here.

% you can choose not to have a title for an appendix
% if you want by leaving the argument blank
%\section{}
%Appendix two text goes here.


% use section* for acknowledgment
%\section*{Acknowledgment}


%The authors would like to thank...


% Can use something like this to put references on a page
% by themselves when using endfloat and the captionsoff option.
\ifCLASSOPTIONcaptionsoff
  \newpage
\fi



% trigger a \newpage just before the given reference
% number - used to balance the columns on the last page
% adjust value as needed - may need to be readjusted if
% the document is modified later
%\IEEEtriggeratref{8}
% The "triggered" command can be changed if desired:
%\IEEEtriggercmd{\enlargethispage{-5in}}

% references section

% can use a bibliography generated by BibTeX as a .bbl file
% BibTeX documentation can be easily obtained at:
% http://mirror.ctan.org/biblio/bibtex/contrib/doc/
% The IEEEtran BibTeX style support page is at:
% http://www.michaelshell.org/tex/ieeetran/bibtex/
%\bibliographystyle{IEEEtran}
% argument is your BibTeX string definitions and bibliography database(s)
%\bibliography{IEEEabrv,../bib/paper}
%
% <OR> manually copy in the resultant .bbl file
% set second argument of \begin to the number of references
% (used to reserve space for the reference number labels box)


% biography section
% 
% If you have an EPS/PDF photo (graphicx package needed) extra braces are
% needed around the contents of the optional argument to biography to prevent
% the LaTeX parser from getting confused when it sees the complicated
% \includegraphics command within an optional argument. (You could create
% your own custom macro containing the \includegraphics command to make things
% simpler here.)
%\begin{IEEEbiography}[{\includegraphics[width=1in,height=1.25in,clip,keepaspectratio]{mshell}}]{Michael Shell}
% or if you just want to reserve a space for a photo:


%\begin{IEEEbiography}{Yuki Okafuji}
%Biography text here.
%\end{IEEEbiography}

%\begin{IEEEbiography}{Yasunori Ozaki}
%Biography text here.
%\end{IEEEbiography}

%\begin{IEEEbiography}{Jun Baba}
%Biography text here.
%\end{IEEEbiography}

%\begin{IEEEbiography}{Junya Nakanishi}
%Biography text here.
%\end{IEEEbiography}

%\begin{IEEEbiography}{Kohei Ogawa}
%Biography text here.
%\end{IEEEbiography}

%\begin{IEEEbiography}{Yuichiro Yoshikawa}
%Biography text here.
%\end{IEEEbiography}

%\begin{IEEEbiography}{Hiroshi Ishiguro}
%Biography text here.
%\end{IEEEbiography}

% if you will not have a photo at all:
%\begin{IEEEbiographynophoto}{John Doe}
%Biography text here.
%\end{IEEEbiographynophoto}

% insert where needed to balance the two columns on the last page with
% biographies
%\newpage

%\begin{IEEEbiographynophoto}{Jane Doe}
%Biography text here.
%\end{IEEEbiographynophoto}

% You can push biographies down or up by placing
% a \vfill before or after them. The appropriate
% use of \vfill depends on what kind of text is
% on the last page and whether or not the columns
% are being equalized.

%\vfill

% Can be used to pull up biographies so that the bottom of the last one
% is flush with the other column.
%\enlargethispage{-5in}



% that's all folks
\end{document}


